Author's response to reviews

Title: Asthma in changing environments - chances and challenges of international research collaborations between South America and Europe - Study protocol and first results of a case-control study

Authors:

Anja Boneberger (Anja.Boneberger@med.uni-muenchen.de)
Katja Radon (sekretariat-radon@lmu.de)
Jennifer Baer (Jennifer.C.Baer@web.de)
Leonie Kausel (leoniekausel@gmail.com)
Michael Kabesch (Kabesch.Michael@mh-hannover.de)
Daniel Haider (Haider.Daniel@gmx.de)
Rudolf Schierl (Rudolf.Schierl@med.uni-muenchen.de)
Rüdiger von Kries (Ruediger.Kries@med.uni-muenchen.de)
Mario Calvo (mcalvo@uach.cl)

Version: 2 Date: 4 August 2010

Author's response to reviews:

Abstract:
Discussion: Overall 1,173 participants were contacted. Response was 82% among cases and 65% among controls. Atopic sensitization was high (78% among cases, 47% among controls).

Figure 2 does not exist. Figure 2 = Figur1